<%namespace name="tools" file="tools.tex"/>
\textbf{Results morphology}\newline
The comparison between the observed and numerically predicted beach profiles is shown in \autoref{fig:T01_profile},  \autoref{fig:T02_profile} and \autoref{fig:T03_profile}. The observed profiles are represented by a dashed black line and the computed profiles through a solid line (upper panel red, lower panel gray). The dune erosion, foreshore slope and dune retreat are shown in \autoref{tab:DG2006_overview}. 

${tools.image('Deltaflume2006_nh/T01/', 'T01_profile2.png', 'Profile development during test T01. The upper panel shows the initial bel level (black), computed erosion bed level (red) and observed bed level (dashed black) together with a water level snapshot. The lower panel shows the observed (dashed black line) and modelled (gray line) the bed level change caused erosion/decomposition.',label='T01_profile')}
${tools.image('Deltaflume2006_nh/T02/', 'T02_profile2.png', 'Profile development during test T02. The upper panel shows the initial bel level (black), computed erosion bed level (red) and observed bed level (dashed black) together with a water level snapshot. The lower panel shows the observed (dashed black line) and modelled (gray line) the bed level change caused erosion/decomposition.',label='T02_profile')}
${tools.image('Deltaflume2006_nh/T03/', 'T03_profile2.png', 'Profile development during test T03. The upper panel shows the initial bel level (black), computed erosion bed level (red) and observed bed level (dashed black) together with a water level snapshot. The lower panel shows the observed (dashed black line) and modelled (gray line) the bed level change caused erosion/decomposition.',label='T03_profile')}

${tools.table('Deltaflume2006_nh/', 'skill.tex', 'Statistical scores for Deltagoot 2006 experiments', 'DG2006_overview')} 

\textbf{Results hydrodynamics}\newline
Next to the morphology also the hydrodynamics are compared to the observations. The hydrodynamics of the experiments are shown in \autoref{fig:T01_hydro}, \autoref{fig:T02_hydro} and \autoref{fig:T03_hydro}.

${tools.image('Deltaflume2006_nh/T01/', 'T01_hydro.png', 'Hydrodynamics during test T01. The upper panel shows the erosion profile. The wave height is shown in the second panel where the observed wave height is shown with black dots and the modeled wave height with the solid line. The third panel shows the wave setup where the observed wave height is shown with black dots and the modeled wave height with the solid line',label='T03_hydro')}
${tools.image('Deltaflume2006_nh/T02/', 'T02_hydro.png', 'Hydrodynamics during test T02. The wave height is shown in the second panel where the observed wave height is shown with black dots and the modeled wave height with the solid line. The third panel shows the wave setup where the observed wave height is shown with black dots and the modeled wave height with the solid line',label='T03_hydro')}
${tools.image('Deltaflume2006_nh/T03/', 'T03_hydro.png', 'Hydrodynamics during test T03. The wave height is shown in the second panel where the observed wave height is shown with black dots and the modeled wave height with the solid line. The third panel shows the wave setup where the observed wave height is shown with black dots and the modeled wave height with the solid line',label='T03_hydro')}
